A Phase I trial of talazoparib in patients with advanced hematologic malignancies

Ajay K Gopal^{*,1}, Rakesh Popat², Ryan J Mattison³, Tobias Menne⁴, Adrian Bloor⁵, Terry Gaymes⁶, Asim Khwaja⁷, Mark Juckett³, Ying Chen⁸, Matthew J Cotter⁸ & Ghulam J Mufti⁹

¹Division of Medical Oncology, Department of Medicine, University of Washington, Seattle, Clinical Research Division, Fred Hutchinson Cancer Research Center, Seattle, WA 98109, USA

²National Institute for Health Research University College London Hospitals Clinical Research Facility, University College London Hospitals NHS Foundation Trust, London, UK

³Carbone Cancer Center, University of Wisconsin, Madison, WI, USA

⁴Department of Hematology, Freeman Hospital, Newcastle Upon Tyne Hospitals NHS Foundation Trust, Newcastle Upon Tyne, UK

⁵The Christie NHS Foundation Trust, University of Manchester, Manchester, UK

⁶Department of Biomolecular Science, Kingston University, London, UK

⁷University College London Cancer Institute & University College London Hospitals NHS Foundation Trust, London, UK

⁸Pfizer Inc., New York, NY, USA

⁹Department of Hematology, King's College London, King's College Hospital NHS Foundation Trust, London, UK

*Author for correspondence: agopal@u.washington.edu

Aim: The objective of this study was to establish the maximum tolerated dose (MTD), safety, pharmacokinetics, and anti-leukemic activity of talazoparib. Patients & methods: This Phase I, two-cohort, dose-escalation trial evaluated talazoparib monotherapy in advanced hematologic malignancies (cohort 1: acute myeloid leukemia/myelodysplastic syndrome; cohort 2: chronic lymphocytic leukemia/mantle cell lymphoma). Results: Thirty-three (cohort 1: n = 25; cohort 2: n = 8) patients received talazoparib (0.1–2.0 mg once daily). The MTD was exceeded at 2.0 mg/day in cohort 1 and at 0.9 mg/day in cohort 2. Grade \geq 3 adverse events were primarily hematologic. Eighteen (54.5%) patients reported stable disease. Conclusion: Talazoparib is relatively well tolerated in hematologic malignancies, with a similar MTD as in solid tumors, and shows preliminary anti leukemic activity.

Clinical trial registration: NCT01399840 (ClinicalTrials.gov)

Lay abstract: The objective of this study was to define the highest dose of talazoparib that people with various types of leukemia (mainly various blood cancers) could tolerate. People were assigned into two cohorts based on their type of leukemia: cohort 1 included 25 people with acute myeloid leukemia or myelodysplastic syndrome; cohort 2 included 8 people with chronic lymphocytic leukemia or mantle cell lymphoma. Similar to what researchers observed for people with solid tumors, the highest tolerated dose was 1.35 mg per day in cohort 1, and it was estimated to be \sim 0.9 mg per day in cohort 2. Side effects that occurred during the study were expected, given the types of leukemia being treated. Talazoparib also showed promising anti leukemic effects in some patients.

Tweetable abstract: In this Phase I talazoparib trial in hematologic malignancies (cohort 1: AML/MDS, n = 25; cohort 2: CLL/MCL, n = 8), the maximum tolerated dose was exceeded at 2.0 and 0.9 mg/day in cohorts 1 and 2, respectively. Stable disease and transfusion independence were also observed.

First draft submitted: 5 April 2021; Accepted for publication: 30 July 2021; Published online: 23 October 2021

Keywords: *BRCA1/2* mutations • DNA damage response • hematologic malignancy • poly(ADP-ribose) polymerase inhibition • talazoparib

Introduction

Hematologic malignancies are heterogeneous blood cancers that vary in incidence, etiology, and survival [1]. Because of population growth and increased longevity, there has been an increase in the global incidence of hematologic



malignancies in recent years [2]. Indeed, between 2006 and 2016, the global incidence of leukemia and non-Hodgkin lymphoma increased by 26% (370,482 to 466,802 cases) and 45% (319,078 to 461,164 cases), respectively [2].

Similar to most other cancers, hematologic malignancies may exhibit deficiencies in DNA damage response (DDR) pathways, including homologous recombination, that render cells highly dependent on alternative DNA repair mechanisms such as those mediated by PARP enzymes 1 and 2 [3,4]. There is substantial clinical evidence demonstrating that PARP inhibitors can effectively treat solid tumors with DDR deficiencies, whereby tumor cells with *BRCA1/2* mutations are selectively killed via synthetic lethality [5,6]. Although clinical evidence for PARP inhibition in hematologic malignancies is limited, there are considerable pre-clinical data pointing to an association between hematologic cancers and several DDR gene mutations. For example, alterations in *ATM* have been implicated in mantle cell lymphoma (MCL), chronic lymphocytic leukemia (CLL), and myelodysplastic syndrome (MDS) [4,7]. Additionally, mutations in *CHEK2* and *BRCA2* have been linked to CLL and MDS [4,8,9], *RAD51* in acute myeloid leukemia (AML) and MDS [10,11], *MLH1* in MDS [12,13], and *BRCA1* in AML [4]. Moreover, both *in vitro* and *in vivo* studies have demonstrated that ATM-deficient lymphoid tumor cells, i.e., CLL and MCL, are sensitive to PARP inhibition, with data showing cytotoxicity as well as reduced tumor growth and prolonged survival in animal models [14–17]. In addition, low *BRCA1* mRNA expression has been associated with sensitivity to PARP inhibition, both in CLL and AML [18,19].

The PARP inhibitor talazoparib has demonstrated efficacy as a monotherapy in cancers with germline *BRCA1/2* mutations [20–22], and it has also shown higher PARP-trapping activity *in vitro* compared with other PARP inhibitors [3,23,24]. A Phase I, dose-escalation trial of talazoparib in patients with germline *BRCA1/2* mutations and solid tumors (NCT01286987) established the maximum tolerated dose (MTD) as 1 mg once daily [22]. In the Phase III EMBRACA trial (NCT01945775) that followed, patients with germline *BRCA1/2*-mutated HER2- advanced breast cancer treated with 1 mg once daily talazoparib had significantly longer progression-free survival and clinically meaningful improvements in patient-reported outcomes compared with physician's choice of chemotherapy [20], leading to regulatory approvals for this indication in the US, EU, and multiple other countries [25,26]. In addition, talazoparib is in clinical development for metastatic castration-resistant prostate cancer (mCRPC), including as monotherapy in previously treated patients with DDR-deficient mCRPC (NCT03148795) [27], and as a first-line treatment in combination with enzalutamide in molecularly unselected patients with mCRPC (NCT03395197) [28].

This Phase I, two-cohort, dose-escalation trial of talazoparib monotherapy in molecularly unselected patients with advanced hematologic malignancies (ClinicalTrials.gov Identifier: NCT01399840) was conducted prior to the approval of talazoparib to treat *BRCA1/2*-mutated HER2- advanced breast cancer [25,26] and in parallel with the solid tumor Phase I study (NCT01286987) that determined the talazoparib MTD at 1 mg once daily [22]. Given the pre-clinical rationale for the use of PARP inhibitors in hematologic malignancies, it was of interest to evaluate the MTD, safety, and anti leukemic effects of talazoparib in this hematology-specific Phase I study. The primary objective of this study was to establish the MTD of talazoparib in patients with AML, MDS, CLL, or MCL, with secondary objectives including the evaluation of the safety, pharmacokinetics (PK), and preliminary anti-leukemic activity of talazoparib in these disorders.

Patients & methods

Study population

The study was divided into two cohorts as it was anticipated that impaired hematopoiesis and, therefore, the toxicity profile and tolerability, may vary between different hematologic malignancies: cohort 1 of the study enrolled patients with AML (relapsed, refractory, or newly diagnosed by WHO's classification [>20% myeloblasts]) or patients with MDS who had failed or declined standard-of-care therapy, while cohort 2 included patients with CLL or MCL, who had relapsed, were refractory or intolerant of standard treatment, or had declined standard therapy.

Patients enrolled in the study were ≥ 18 years of age, with an Eastern Cooperative Oncology Group (ECOG) performance status ≤ 1 . Other inclusion criteria included adequate organ function, defined as serum AST and ALT values $\leq 2.5 \times$ upper limit of normal (ULN) and total serum bilirubin $\leq 1.5 \times$ ULN ($\leq 3 \times$ ULN for Gilbert's syndrome). Patients also needed to demonstrate recovery from acute toxicity due to prior treatment. Key exclusion criteria included acute promyelocytic leukemia and certain disease-specific criteria; these included marrow cellularity <25% or circulating blasts $>50 \times 10^9$ /L for AML, and platelet count $<50 \times 10^9$ /L and neutrophil count $<1 \times 10^9$ /L for MCL/CLL (unless decreased counts were secondary to bone marrow effacement due to leukemia or lymphoma, splenomegaly, or autoimmune thrombocytopenia). Additional key exclusion criteria included prior allogeneic or autologous bone marrow transplant <6 months before cycle 1 day 1 and/or the

presence of graft-versus-host disease (acute or chronic; allogeneic bone marrow transplant only). Exclusionary prior treatment included anti-leukemia treatment within 14 days or hydroxyurea treatment within 7 days of first dose for AML; anti-lymphoma/leukemia treatment within 28 days of first dose for CLL, MCL, or MDS patients; and transfusion/hematopoietic growth factors within 7 days of first dose for CLL and MCL patients.

Study design

This study was an open-label, two-cohort, Phase I, dose-escalation study of once-daily talazoparib monotherapy in patients with advanced hematologic malignancies (ClinicalTrials.gov Identifier: NCT01399840) conducted at seven sites in the US and the UK between July 2011 and November 2013. Enrollment and dose escalation were conducted independently in each cohort of the study. Talazoparib was administered orally, once a day for repeated 21-day cycles under fasting conditions. Each cohort consisted of two stages: dose escalation (stage 1) and dose expansion (stage 2). Dose escalation followed a standard 3 + 3 dose-escalation design [29], where the first patient in each group was observed from cycle 1 day 1 through cycle 1 day 8 before two to five additional patients were treated in that dose level. Groups of three to six patients were enrolled in each cohort to receive increasing talazoparib doses; each dose level had to be expanded to six patients if a dose-limiting toxicity (DLT) was observed in one patient. Planned talazoparib dose levels included 0.1, 0.2, 0.3, 0.45, 0.9, 1.35, and 2.0 mg/day.

Study endpoints

The primary objective of the study was to establish the MTD of daily oral talazoparib in molecularly unselected patients with AML or MDS (cohort 1) and in patients with CLL or MCL (cohort 2). The MTD was defined as the highest dose level at which no more than one of six patients experienced a DLT. A DLT was defined as any of the following treatment-related toxicities occurring during cycle 1: in cohort 1, hematologic adverse events (AEs) included bone marrow hypoplasia ($\leq 10\%$ cellularity) lasting ≥ 28 days. Non-hematologic toxicities included any Grade \geq 3 AE except for the following: (1) a non-hematologic Grade 3 laboratory AE that is asymptomatic and rapidly reversible (returned to baseline or to Grade ≤ 1 within 3 days) unless identified as clinically relevant by the investigator; (2) Grade 3 nausea, vomiting, and/or diarrhea responsive to medications within 24 h; (3) Grade 3 tumor lysis syndrome; (4) Grade 3 stomatitis that resolves within 7 days; or (5) Grade 3 fatigue unless ≥ 2 grade increase from baseline lasting >3 days. Missing >4 doses of talazoparib during cycle 1 for treatment-related nonhematologic toxicity was also defined as a DLT. In cohort 2, hematologic AEs included Grade 4 thrombocytopenia persisting for >7 days and/or resulting in Grade ≥ 2 hemorrhage, or Grade 4 neutropenia lasting >7 days and/or associated with fever (hematologic AEs for patients with bone marrow effacement [i.e., platelet count $<50 \times 10^9$ /L and neutrophil count $<1 \times 10^9/L$] included >50% decrease in platelet count persisting for >7 days or resulting in Grade ≥ 2 hemorrhage, or >50% decrease in neutrophil count and/or associated with fever or systemic infection). For non-hematologic AEs, the same criteria used for cohort 1 were used to define DLT.

AEs were coded in accordance with The Medical Dictionary for Regulatory Activities, version 16.1. Treatmentemergent AEs (TEAEs) were defined as AEs that were new, increased in frequency, or worsened in severity, after the first talazoparib dose. AEs counted as drug-related had been classified by investigators as possibly or probably related to talazoparib.

The secondary objectives of the study were to assess the safety, PK, and preliminary anti-leukemic activity of talazoparib in cohorts 1 and 2. Objective response, duration of objective response, and time to progression or treatment failure were recorded. Response was measured by the following response criteria: the European leukemiaNet for subjects with AML [30], the International Working Group for Prognosis in MDS for subjects with MDS [31,32], the International Workshop on Chronic Lymphocytic leukemia (iwCLL) [33] for subjects with CLL, and the Revised Response Criteria for MCL [34].

Dose modifications

Patients experiencing a DLT during cycle 1 could resume talazoparib treatment at the same or reduced dose after recovery to baseline or Grade ≤ 1 within 21 days of treatment interruption. Patients not experiencing a DLT during cycle 1 could receive additional treatment until disease progression, treatment failure, or relapse. Patients with persisting non-hematologic Grade 1/2 toxicity were dose reduced ~25%. Dosing was stopped for patients with a Grade 3/4 non-hematologic toxicity and resumed with a 25%–50% dose reduction if toxicity resolved to Grade 0/1 after >7 days, or returned to baseline. Patients with non-hematologic Grade 3 toxicity that resolved to Grade 0/1 in <7 days could resume with ~25% reduction.





AML: Acute myeloid leukemia; CLL: Chronic lymphocytic leukemia; MCL: Mantle cell lymphoma; MDS: Myelodysplastic syndrome.

PK sample collection & assessments

Plasma and urine samples were assayed for talazoparib concentrations by Covance Laboratories Inc., using validated high performance liquid chromatography with tandem mass spectrometry detection methods [22]. The lower limits of quantitation for serum and urine were 5.0 pg/ml and 25.0 pg/ml, respectively. Evaluated PK parameters included maximum concentration (C_{max}), minimum concentration (C_{min}), time to maximum plasma concentration (T_{max}), area under the curve from time 0 to last measurable concentration (AUC_{0-t}), area under the curve from time 0 to 24 h post-dose (AUC₀₋₂₄), amount of drug excreted into the urine from time 0 to 24 h (Ae₀₋₂₄), fraction of administered drug excreted in the urine from time 0 to 24 h (Fe₀₋₂₄), and average renal clearance from time 0 to 24 h (ARC₀₋₂₄); ARC₀₋₂₄ was calculated as (Ae₀₋₂₄/AUC₀₋₂₄). PK parameters were calculated using PhoenixTM WinNonlin[®] 6.1.

Statistical analyses

All analyses in this open-label, dose-finding study were descriptive in nature. Descriptive summaries of continuous variables, including the mean, standard deviation (SD), median, range, and, where appropriate, 95% CI and/or interquartile range, were determined using the Statistical Analysis Software (SAS), version 9.2. Descriptive summaries of categorical variables included the number of patients and percentages. Data were summarized with respect to the dose level at which the patients started treatment. Cohorts were defined as all patients who began treatment at a given dose level.

For determination of the MTD, only the treated evaluable patients were included. No formal sample size calculations based on statistical power were performed. Descriptive statistics were used to characterize all outcomes.

Summary statistics for plasma and urine talazoparib concentrations and PK parameters were calculated by grouping subjects in cohorts 1 and 2 together to provide a larger sample size at each dose level.

Results

Patients

A total of 33 patients (23 males, ten females) were treated with talazoparib during the study and included in the safety analyses. There were 25 patients in cohort 1 (21 with AML, four with MDS) and eight patients in cohort 2 (four with CLL, four with MCL; Figure 1, Supplementary Table 1).

The median age of patients was 70.0 and 75.5 years in cohort 1 and cohort 2 of the study, respectively (Table 1). Patients were predominantly males and white (Supplementary Table 1). The median number of previous

Table 1. Participant demographics and baseline clinical characteristics (safety population).											
Dose level, mg/day	Cohort 1									Cohort 2	
	0.10 (n = 3)	0.20 (n = 3)	0.30 (n = 5)	0.45 (n = 3)	0.90 (n = 4)	1.35 (n = 3)	2.00 (n = 4)	All (n = 25)	0.10 (n = 3)	0.90 (n = 5)	All (n = 8)
Median age (range), years	56.0 (37, 69)	79.0 (72, 83)	71.0 (66, 76)	62.0 (33, 71)	67.5 (22, 74)	70.0 (40, 74)	68.5 (65, 71)	70.0 (22, 83)	82.0 (68, 86)	75.0 (49, 76)	75.5 (49, 86)
Sex, n (%) Female Male	1 (33.3) 2 (66.7)	1 (33.3) 2 (66.7)	1 (20.0) 4 (80.0)	1 (33.3) 2 (66.7)	2 (50.0) 2 (50.0)	1 (33.3) 2 (66.7)	1 (25.0) 3 (75.0)	8 (32.0) 17 (68.0)	0 (0.0) 3 (100)	2 (40.0) 3 (60.0)	2 (25.0) 6 (75.0)
Race, n (%) [†] White Asian Black Anglo–Indian Turkish	2 (66.7) 0 (0.0) 0 (0.0) 1 (33.3) 0 (0.0)	3 (100) 0 (0.0) 0 (0.0) 0 (0.0) 0 (0.0)	4 (80.0) 0 (0.0) 0 (0.0) 0 (0.0) 1 (20.0)	3 (100) 0 (0.0) 0 (0.0) 0 (0.0) 0 (0.0)	4 (100) 0 (0.0) 0 (0.0) 0 (0.0) 0 (0.0)	3 (100) 0 (0.0) 0 (0.0) 0 (0.0) 0 (0.0)	4 (100) 0 (0.0) 0 (0.0) 0 (0.0) 0 (0.0)	23 (92.0) 0 (0.0) 0 (0.0) 1 (4.0) 1 (4.0)	3 (100) 0 (0.0) 0 (0.0) 0 (0.0) 0 (0.0)	4 (80.0) 1 (20.0) 0 (0.0) 0 (0.0) 0 (0.0)	7 (87.5) 1 (12.5) 0 (0.0) 0 (0.0) 0 (0.0)
BMI, kg/m² n Median (range)	3 24.9 (20.6, 27.9)	2 21.1 (19.8, 22.4)	3 28.4 (24.8, 35.9)	3 30.8 (28.8, 47.2)	4 24.7 (21.1, 33.3)	3 25.8 (24.1, 29.4)	4 27.8 (25.0, 29.5)	22 27.3 (19.8, 47.2)	3 27.2 (19.7, 29.0)	5 23.3 (18.3, 32.9)	8 24.4 (18.3, 32.9)
ECOG, n (%) [‡] 0 1	1 (33.3) 2 (66.7)	0 (0.0) 3 (100)	2 (40.0) 3 (60.0)	0 (0.0) 3 (100)	0 (0.0) 4 (100)	1 (33.3) 2 (66.7)	1 (25.0) 3 (75.0)	5 (20.0) 20 (80.0)	1 (33.3) 2 (66.7)	4 (80.0) 1 (20.0)	5 (62.5) 3 (37.5)

[†]None of the patients were of Hispanic ethnicity.

[‡]ECOG performance status 0 = fully active, able to carry on all pre-disease performance without restriction; ECOG performance status 1 = restricted in physically strenuous activity but ambulatory and able to carry out work of a light or sedentary nature.

ECOG: Eastern Cooperative Oncology Group.

Table 2. Dose-limiting toxicity adverse events (safety population).												
Dose level, mg/day	Cohort 1									Cohort 2		
	0.10 (n = 3)	0.20 (n = 3)	0.30 (n = 5)	0.45 (n = 3)	0.90 (n = 4)	1.35 (n = 3)	2.00 (n = 4)	All (n = 25)	0.10 (n = 3)	0.90 (n = 5)	All (n = 8)	
Any DLT AE, n (%)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	2 (50.0)	2 (8.0)	0 (0.0)	2 (40.0)	2 (25.0)	
Blood and lymphatic system disorders, n (%) Febrile neutropenia	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	1 (25.0)	1 (4.0)	0 (0.0)	0 (0.0)	0 (0.0)	
Neutropenia	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	2 (40.0)	2 (25.0)	
Infections and infestations, n (%) Neutropenic sepsis	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	1 (25.0) [†]	1 (4.0)	0 (0.0)	0 (0.0)	0 (0.0)	

[†]One death due to neutropenic sepsis, which occurred in a patient with AML who received the highest dose of 2 mg talazoparib, was considered to be possibly drug-related by the investigator; further details on the patient's medical history can be found in the Supplementary Material. AE: Adverse event; AML: Acute myeloid leukemia; DLT: Dose-limiting toxicity.

chemotherapy regimens was three (maximum = seven) and six (maximum = 13) in cohorts 1 and 2, respectively.

All patients completed the study as required by the protocol. Following study completion, primary reasons for ending treatment included disease progression (69.7% [n = 23]), withdrawal by subject (3.0% [n = 1]), physician decision (12.1% [n = 4]), withdrawal due to an AE (3.0% [n = 1]) and death (12.1% [n = 4]; Figure 1).

Exposure to talazoparib

The median talazoparib exposure duration for patients in cohort 1 and cohort 2 of the study was 49.0 days (range: 7.0–484.0) and 34.5 days (range: 13.0–202.0), respectively (Supplementary Table 2). The mean daily dose of talazoparib administered to patients in cohort 1 and cohort 2 of the study was 0.74 mg (SD: 0.61 mg) and 0.52 mg (SD: 0.39 mg), respectively (Supplementary Table 2).

Maximum tolerated dose

Within cohort 1, all seven dose groups (0.1–2.0 mg/day) were explored, with four patients (three with AML, one with MDS) receiving the highest dose of 2.0 mg/day (Supplementary Table 1). DLTs in cycle 1 were observed in two patients (Table 2): at the 2.0 mg dose, one patient with MDS experienced febrile neutropenia and one patient with AML experienced neutropenic sepsis. The MTD was determined as 1.35 mg/day as no DLTs were observed in the three patients treated with this dose level.

Table 3. Summary of treatment-emergent adverse events with talazoparib (safety population). †											
			Cohort 2								
Dose level, mg/day	0.10 (n = 3)	0.20 (n = 3)	0.30 (n = 5)	0.45 (n = 3)	0.90 (n = 4)	1.35 (n = 3)	2.00 (n = 4)	All (n = 25)	0.10 (n = 3)	0.90 (n = 5)	All (n = 8)
Any AEs, n (%)	3 (100)	3 (100)	5 (100)	3 (100)	4 (100)	3 (100)	4 (100)	25 (100)	3 (100)	5 (100)	8 (100)
Any DLT AEs	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	2 (50.0)	2 (8.0)	0 (0.0)	2 (40.0)	2 (25.0)
Any study drug-related AEs \ddagger	3 (100)	1 (33.3)	4 (80.0)	1 (33.3)	3 (75.0)	3 (100)	2 (50.0)	17 (68.0)	1 (33.0)	5 (100)	6 (75.0)
Any AEs of CTCAE Grade 3 or 4, n (%)	2 (66.7)	2 (66.7)	5 (100)	3 (100)	4 (100)	3 (100)	3 (75.0)	22 (88.0)	3 (100)	3 (60.0)	6 (75.0)
Any study drug-related AEs of CTCAE Grade of 3 or 4^{\ddagger}	1 (33.3)	0 (0.0)	3 (60.0)	0 (0.0)	2 (50.0)	2 (66.7)	2 (50.0)	10 (40.0)	1 (33.3)	3 (60.0)	4 (50.0)
Any AEs of CTCAE Grade of 5 (death), n (%)	0 (0.0)	2 (66.7)	1 (20.0)	2 (66.7)	3 (75.0)	1 (33.3)	3 (75.0)	12 (48.0)	1 (33.3)	0 (0.0)	1 (12.5)
Any study drug-related AEs of CTCAE Grade 5 (death) ‡	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	1 (25.0) [§]	1 (4.0) [§]	0 (0.0)	0 (0.0)	0 (0.0)
Any SAEs, n (%)	2 (66.7)	2 (66.7)	4 (80.0)	3 (100)	4 (100)	2 (66.7)	4 (100)	21 (84.0)	2 (66.7)	2 (40.0)	4 (50.0)
Any study drug-related ${\rm SAEs}^\ddagger$	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	1 (25.0)	0 (0.0)	2 (50.0)	3 (12.0)	0 (0.0)	0 (0.0)	0 (0.0)
Any SAEs of CTCAE Grade of 3 or 4	2 (66.7)	1 (33.3)	4 (80.0)	2 (66.7)	2 (50.0)	2 (66.7)	2 (50.0)	15 (60.0)	1 (33.3)	1 (20.0)	2 (25.0)
Any AEs leading to dose interruption or modification, n (%)	1 (33.3)	2 (66.7)	4 (80.0)	0 (0.0)	3 (75.0)	0 (0.0)	2 (50.0)	12 (48.0)	0 (0.0)	3 (60.0)	3 (37.5)
Any study drug-related AEs leading to dose interruption or modification [‡]	1 (33.3)	1 (33.3)	2 (40.0)	0 (0.0)	1 (25.0)	0 (0.0)	2 (50.0)	7 (28.0)	0 (0.0)	3 (60.0)	3 (37.5)
Any AEs leading to study drug withdrawn, n (%)	0 (0.0)	1 (33.3)	1 (20.0)	0 (0.0)	0 (0.0)	0 (0.0)	1 (25.0)	3 (12.0)	0 (0.0)	0 (0.0)	0 (0.0)
Any study drug-related AEs leading to study drug withdrawal [‡]	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	1 (25.0)	1 (4.0)	0 (0.0)	0 (0.0)	0 (0.0)

[†]Adverse events were coded with MedDRA version 16.1.

[‡]AEs that were classified by the investigator as possibly or probably related to the study drug.

[§]One death due to neutropenic sepsis, which occurred in a patient with AML who received the highest dose of 2 mg talazoparib, was considered to be possibly drug-related by the investigator; further details on the patient's medical history can be found in the Supplementary Material.

AE: Adverse event; AML: Acute myeloid leukemia; CTCAE: Common Terminology Criteria for Adverse Events; DLT: Dose-limiting toxicity; MedDRA: The Medical Dictionary for Regulatory Activities; SAE: Serious adverse event.

Within cohort 2, dose levels of 0.2, 0.3, and 0.45 mg/day were omitted because this dose range was well tolerated in the parallel dose-escalation study in patients with solid tumors [22]. As a result, only two dose groups (0.1 and 0.9 mg/day) were enrolled and treated, with five patients (three with CLL and two with MCL) receiving the maximum dose of 0.9 mg/day (Supplementary Table 1). DLTs in cycle 1 were observed in two patients (Table 2): at 0.9 mg/day, one patient with MCL and one patient with CLL experienced Grade 4 neutropenia. As no intermediate doses were investigated, the MTD was not further elucidated but was concluded to be exceeded at 0.9 mg/day in cohort 2 of the study.

Adverse events

All patients experienced ≥ 1 TEAE (Table 3). When both cohorts were considered collectively, the most frequently reported TEAEs were fatigue, pyrexia, and extremity pain (Table 4). TEAEs considered by investigators to be possibly or probably related to talazoparib (drug-related) were reported for 23 (69.7%) patients (Table 3); among these, the most frequently reported TEAEs were fatigue, neutropenia, and nausea (Table 4).

Of the Grade 3/4 TEAEs reported for 28 (84.8%) patients, the most frequently reported were neutropenia, thrombocytopenia, febrile neutropenia, and anemia (Table 3-4). Drug-related Grade 3/4 TEAEs were reported for 14 (42.4%) patients (Table 3). The most frequently reported drug-related Grade 3/4 TEAEs were neutropenia and thrombocytopenia (Table 4).

Serious AEs (SAEs) were reported for 25 (75.8%) patients, of which 17 (51.5%) patients had a Grade 3/4 SAE and 3 (9.1%) patients had a drug-related SAE (two patients reported neutropenic sepsis [grade 3 and 5, respectively], one patient reported shortness of breath [grade 2] and neutropenic fever [grade 4]; Table 3). SAEs in the infections and infestations system organ class were the most common in both cohorts (including neutropenic sepsis, pneumonia,

Table 4. Most frequently reported treatment-emergent adverse events with talazoparib (safety population). [†]									
n = 33	Grade 1–2	Grade 3–4	All grades [‡]						
Most frequently reported TEAEs, \S n (%)									
Fatigue	10 (30.3)	3 (9.1)	13 (39.4)						
Pyrexia	12 (36.4)	0 (0.0)	12 (36.4)						
Pain in extremity	10 (30.3)	1 (3.0)	11 (33.3)						
Dyspnea	8 (24.2)	2 (6.1)	10 (30.3)						
Epistaxis	9 (27.3)	1 (3.0)	10 (30.3)						
Febrile neutropenia	2 (6.1)	8 (24.2)	10 (30.3)						
Cough	9 (27.3)	0 (0.0)	9 (27.3)						
Diarrhea	9 (27.3)	0 (0.0)	9 (27.3)						
Neutropenia	0 (0.0)	9 (27.3)	9 (27.3)						
Thrombocytopenia	0 (0.0)	9 (27.3)	9 (27.3)						
Vomiting	9 (27.3)	0 (0.0)	9 (27.3)						
Anemia	0 (0.0)	8 (24.2)	8 (24.2)						
Nausea	8 (24.2)	0 (0.0)	8 (24.2)						
Abdominal pain	7 (21.2)	0 (0.0)	7 (21.2)						
Back pain	6 (18.2)	1 (3.0)	7 (21.2)						
Chills	7 (21.2)	0 (0.0)	7 (21.2)						
Decreased appetite	7 (21.2)	0 (0.0)	7 (21.2)						
Hypokalemia	5 (15.2)	2 (6.1)	7 (21.2)						
Pneumonia	4 (12.1)	2 (6.1)	6 (18.2)						
Most frequently reported drug-related AEs, [¶] n (%)									
Fatigue	8 (24.2)	1 (3.0)	9 (27.3)						
Neutropenia	0 (0.0)	9 (27.3)	9 (27.3)						
Nausea	8 (24.2)	0 (0.0)	8 (24.2)						
Thrombocytopenia	0 (0.0)	4 (12.1)	4 (12.1)						
Neutropenic sepsis [#]	0 (0.0)	1 (3.0)	1 (3.0)						

 $^\dagger \text{Adverse}$ events were coded with MedDRA, version 16.1.

[‡]All Grades include Grade 1–4 events only; Grade 5 events include neutropenic sepsis (possibly drug-related; further details below) and pneumonia (due to progressive disease).

§ Adverse events include those preferred terms reported for at least seven (21.2%) subjects in the study (N = 33). Subjects with more than one AE within a MedDRA system organ class and preferred term were counted once for the worst grade within that system organ class and preferred term.

[¶]Adverse events include those preferred terms reported for at least four (12.1%) subjects and the single subject with a drug-related death (Grade 5). Subjects with more than one AE within a MedDRA system organ class and preferred term were counted once for the worst grade within that system organ class and preferred term.

[#]There was one Grade 5 drug-related event (1/33 patients, 3.0%; not including deaths related to disease progression), which included neutropenic sepsis reported in a patient with AML who received the highest dose of 2 mg talazoparib; this was considered to be possibly drug-related by the investigator. Further details on the patient's medical history can be found in the Supplementary Material.

AE: Adverse event; AML: Acute myeloid leukemia; MedDRA: The Medical Dictionary for Regulatory Activities; TEAE: Treatment-emergent adverse event.

respiratory tract infection, and bacteremia). At least one TEAE leading to dose interruption and/or reduction was observed for 15 patients (45.5%). For ten patients (30.3%), TEAEs leading to dose interruption/reduction were reported to be related to study drug, and were primarily hematologic in nature (Table 3).

Deaths

Thirteen (39.4%) deaths were reported (Table 3; three patients with AML and one with MDS died during the study; eight patients with AML and one with MCL died of disease progression within 30 days after the last study drug dose). All deaths were either related to the advanced nature of underlying hematologic malignancies impairing hematologic and immunologic functions (10/13 patients), or to consequent infection, febrile neutropenia, or neutropenic sepsis (3/13 patients). One death due to neutropenic sepsis, which occurred in a patient with AML who received the highest dose of 2 mg talazoparib, was considered to be possibly drug-related by the investigator. This patient had a history of neutropenia and thrombocytopenia and, at study baseline, had Grade 4 leukopenia, neutropenia, and thrombocytopenia; further details on the patient's medical history can be found in the Supplementary Material.

Table 5. Pharmacokinetic parameters in plasma and urine after multiple daily talazoparib dosing.										
	Parameter summary statistics [†] by dose, mg/day									
	0.10	0.20	0.30	0.45	0.90	1.35	2.00			
Plasma PK	n = 6 [‡]	n = 3	n = 4	n = 2	n = 4	n = 3	n = 2			
T _{max} , hr	1.53 (0.250, 2.02)	0.983 (0.550, 1.03)	1.63 (0.917, 2.98)	4.50 (1.00, 8.00)	1.48 (1.00, 2.00)	1.03 (1.00, 4.00)	1.00 (1.00, 1.00)			
C _{max} , ng/ml	1.280 (30.9)	3.130 (25.8)	3.460 (49.9)	8.090 (59.7)	12.20 (52.2)	19.70 (33.0)	30.50 (114)			
AUC _{0−24} , ng•hr/ml	16.80 (31.2)	48.60 (21.6)	45.50 (32.6)	111.0 (26.6)	142.0 (63.3)	226.0 (16.0)	454.0 (128)			
AUC₀-t, ng∙hr/ml	16.70 (26.7)	49.60 (20.7)	36.50 (26.6)	111.0 (26.6)	138.0 (65.1)	225.0 (15.9)	405.0 (126)			
AR	7.87 (5.55, 8.46)	6.79 (5.76, 7.19)	8.57 (4.56, 14.1)	8.03 (7.57, 8.48)	5.15 (1.92, 16.7)	3.80 (3.28, 4.22)	3.62 (0.490, 6.75)			
Urine PK	n = 4 [§]	n = 2	n = 2	n = 2	n = 5¶	n = 2	n = 2			
Ae ₀₋₂₄ , mg	0.043 (56.4)	0.090 (35.3)	0.162 (30.1)	0.408 (33.7)	0.369 (49.5)	0.531 (29.8)	0.533 (116)			
Fe ₀₋₂₄	0.428 (56.4)	0.450 (35.3)	0.539 (30.1)	0.906 (33.7)	0.410 (49.5)	0.393 (29.8)	0.267 (116)			
ARC ₀₋₂₄ , l/hr	3.51 (57.2)	1.78 (52.4)	4.96 (59.9)	3.64 (7.47)	3.56 (45.8)	2.44 (9.09)	1.66 (45.5)			

 † Geometric mean (geometric CV%) for all parameters except median (range) for T_{max} and AR.

 ‡ n = 4 for AUC₀₋₂₄ and AR.

 $n = 2 \text{ for } ARC_{0-24}$.

 \P n = 4 for ARC₀₋₂₄.

Ae₀₋₂₄: Amount of drug excreted into the urine within the dose interval, 0 to 24 h; AR: Accumulation ratio; ARC₀₋₂₄: Average renal clearance from time 0 to 24 h post-dose; AUC₀₋₂₄: Area under the curve from time 0 to 24 h; AUC₀₋₄: Area under the curve from 0 to last quantifiable sampling point post-dose; C_{max}: Maximum concentration; CV: Coefficient of variation; Fe₀₋₂₄: Fraction of administered drug excreted from time 0 to 24 h; PK: Pharmacokinetics; T_{max}: Time to maximum plasma concentration.

Table 6. Best overall response with stable disease following talazoparib.										
		Cohort 1		Cohort 2						
	AML (n = 21)	MDS (n = 4)	All (n = 25)	CLL (n = 4)	MCL (n = 4)	All (n = 8)				
Stable disease, n (%)	9 (42.9)	4 (100)	13 (52.0)	3 (75.0)	2 (50.0)	5 (62.5)				
Stable disease \geq 16 weeks, [†] n (%)	2 (9.5) [‡]	3 (75.0) [§]	5 (20.0)	0	1 (25.0)	1 (12.5)				

[†]Duration of stable disease is defined for a subject whose best overall response is stable disease as the time from the date of the first dose to the date of the first documentation of progressive disease or treatment failure. If a subject did not have a date of any documentation of progressive disease or treatment failure, the duration of stable disease is defined as the time from the date of the first dose to the date of the last tumor assessment as a censored value.

[‡]One patient became transfusion independent after five treatment cycles.

 ${}^{
m S}$ One patient became transfusion independent after two treatment cycles.

AML: Acute myeloid leukemia; CLL: Chronic lymphocytic leukemia; MCL: Mantle cell lymphoma; MDS: Myelodysplastic syndrome.

PK analysis

Following administration of multiple once-daily doses of talazoparib (cycle 2 day 1), PK analysis showed rapid absorption across the 100–2000 µg dose range and linear PK (Table 5; Supplementary Figure 1). Changes in PK and pharmacodynamics parameters were not specifically analyzed among the relapsed, refractory, and newly diagnosed condition. Results for urinary elimination of the parent compound suggest urinary elimination is the major clearance pathway for talazoparib (Table 5).

Anti leukemic activity

Stable disease was reported for 18 (54.5%) patients. Stable disease of at least 16 weeks' duration was reported in six patients (cohort 1, two patients with AML and three with MDS; cohort 2, one patient with MCL; Table 6). Transfusion independence was reported for two patients, including one with MDS and one with AML.

Discussion

Despite the fact that several pre-clinical studies have demonstrated promising results for PARP inhibitors in hematologic malignancies, there remains an insufficiency of data on the clinical potential of PARP inhibition in these diseases. This Phase I study in hematologic malignancies, conducted between July 2011 and November 2013, aimed to evaluate the MTD of talazoparib monotherapy, as well as provide preliminary results on safety, PK, and antileukemic activity.

Based on the relatively limited, but important, data presented in this study, the MTD in patients with hematologic malignancies appears to be consistent with an MTD of 1 mg once daily reported previously in solid tumors [22].

In cohort 1 of this study, the MTD was exceeded at 2 mg once daily due to dose-limiting febrile neutropenia and fatal neutropenic sepsis, both reported in one patient. The MTD was determined as 1.35 mg once daily because no DLTs were observed in the three patients treated with this dose level. While the MTD was not definitively elucidated in cohort 2 of this study, it was exceeded at 0.9 mg once daily due to dose-limiting Grade 4 neutropenia, which was reported for two patients. As the MTD of talazoparib monotherapy is well established in solid tumors at 1 mg/day in a continuous dosing schedule [22], doses above this, such as those investigated in this study, are likely to have limited clinical applications.

The AEs and DLTs observed in this study were consistent with a study population comprised of patients with advanced hematologic malignancies and potential disease-related hematologic and immunologic dysfunction. The most common Grade 3 and higher AEs were primarily hematologic events (anemia, neutropenia, and thrombocytopenia) or their immediate consequences (e.g., febrile neutropenia). Similarly, the four DLTs reported were also primarily, or related to, hematologic toxicity (neutropenia, febrile neutropenia, and neutropenic sepsis). Overall, the AE profile observed in this study is generally consistent with that reported in solid tumors [22].

Talazoparib demonstrated favorable PK properties with rapid absorption and dose-proportional increases in total exposure over a wide dose range (0.1–2.0 mg/day) after multiple daily doses. These PK results are consistent with those reported previously in solid tumors [22,35].

While single-agent talazoparib demonstrated modest anti leukemic activity in this small, heavily pre-treated, molecularly unselected group of patients, there were promising signals of stable disease and transfusion independence. Indeed, stable disease was reported in over half (18/33) of patients, including two patients with transfusion independence. Disease stability lasting at least 16 weeks was reported in six patients. Overall, these observations suggest that further research of PARP inhibitors in hematologic malignancies may be warranted, particularly to investigate rational combination therapies and identify molecular signatures that may identify patients likely to respond to treatment.

Although molecular signatures are of interest to potentially identify patients likely to benefit from PARP inhibition, there are currently limited clinical data from patients with hematologic malignancies. In a Phase I dose-escalation study in CLL or MCL patients, a trend towards improved overall survival was observed with olaparib in patients with *ATM* mutations versus those without *ATM* mutations; however, this was not statistically significant [36]. It is possible that results from ongoing clinical trials will provide further insight into appropriate patient selection. For example, ongoing trials are assessing the effect of talazoparib monotherapy in cohesin-mutated AML or MDS with excess blasts (NCT03974217), as well as olaparib monotherapy in *IDH*-mutated AML or MDS (ClinicalTrials.gov Identifier: NCT03953898).

As hematologic malignancies represent a heterogeneous group of cancers with complex etiology [1], the combination of PARP inhibitors with other therapies, such as DNA demethylating agents, may increase therapeutic efficacy and overcome potential resistance mechanisms [37]. In AML, talazoparib is currently under investigation in combination with several therapies, including decitabine in untreated, relapsed, or refractory AML (ClinicalTrials.gov Identifier: NCT02878785), as well as with antibody-drug conjugate gemtuzumab ozogamicin in CD33-positive relapsed or refractory AML (NCT04207190). PARP inhibitors enhance response to immunotherapy by promoting neoantigen release, increasing tumor mutational burden and increasing the expression of immune checkpoint regulators such as PD-L1 and CTLA-4 [38]; while promising results have been reported in solid tumors [38], further research in hematologic malignancies may also be relevant.

The results presented here are limited by the variety of molecularly unselected hematologic malignancies, as appropriate for the dose-finding primary objective of this Phase I study, and the genetic signatures of the patient's leukemias are unknown. Additionally, the majority (90.9%) of the patients in the study were white; as there are known racial/ethnic disparities in the incidence and clinical outcomes for patients with hematologic malignancies [39], future studies would benefit by a more inclusive study population. With a clinically diverse population and small patient numbers, however, it is difficult to determine anti leukemic activity. Nevertheless, these limited data suggest that with appropriate patient selection, a clearer signal could be identified.

Conclusion

In conclusion, the AEs and DLTs reported in this study were consistent with the underlying nature of the population's disease. The MTD in patients with hematologic malignancies appears to be in a similar range (\sim 1 mg monotherapy, once daily) to that previously reported in the parallel Phase I study of talazoparib in solid tumors [22]. While limited, single-agent talazoparib in this small, heavily pretreated, molecularly unselected group showed some

promising anti leukemic signals of stable disease and transfusion independence. Therefore, future studies involving PARP inhibitors and patients with hematologic malignancies should continue to investigate rational combination therapies and identify patients likely to benefit from treatment.

Future perspective

Hematologic malignancies are genetically heterogeneous and prone to accumulate DNA damage, providing a therapeutic opportunity for PARP inhibition [40–44]. The use of PARP inhibitors combined with established agents to overcome resistance mechanisms is promising [43]. leukemias including AML, chronic myelogenous leukemia and myeloproliferative disorder have complex mutational landscapes that can evolve over time, and PARP inhibition is an attractive means of exploiting defects in cells that may carry DNA damage response and homologous recombination repair deficiencies due to chromosomal instability [45–48]. PARP inhibitors are currently under investigation for the treatment of these leukemias in combination with decitabine, temozolomide, carboplatin, and topotecan [48]. Combinations with epigenetic modifiers such as histone deacetylase inhibitors may soon follow [48,49].

Future clinical studies are needed to determine if PARP inhibitor combination therapies will be effective in treating patients with hematologic malignancies similar to observations made in solid tumors [5,6]. Clinicians will need to weigh the potential benefits of PARP inhibitors for patients with hematologic malignancies against the potential risks – a recent study showed that patients receiving PARP inhibitors had an increased risk of developing MDS and AML [50]. However, continued study of PARP inhibitors is warranted in patients with hematologic malignancies, particularly as part of a rational combination therapy that may lead to better disease control, especially in patients with poor-risk cytogenetics [48].

Summary points

- This Phase I, two-cohort, dose-escalation trial (ClinicalTrials.gov Identifier: NCT01399840) evaluated talazoparib monotherapy in molecularly unselected patients with advanced hematologic malignancies.
- The primary objective of the study was to establish the maximum tolerated dose (MTD) of talazoparib in patients with acute myeloid leukemia or myelodysplastic syndrome (Cohort 1; n = 25) and in patients with chronic lymphocytic leukemia or mantle cell lymphoma (Cohort 2; n = 8).
- The secondary objectives were to assess the safety, pharmacokinetics, and preliminary anti-leukemic activity of talazoparib in both cohorts.
- Results demonstrated that the MTD was exceeded at 2.0 mg/day in Cohort 1 and at 0.9 mg/day in Cohort 2. Grade ≥3 adverse events were primarily hematologic.
- Stable disease was reported for 18/33 (54.5%) patients, and transfusion independence was achieved in two patients.
- Based on the data presented here, the MTD of talazoparib in patients with hematologic malignancies appears to be consistent with the MTD of 1 mg once daily reported in solid tumors.
- Talazoparib was relatively well tolerated and the AEs observed were consistent with the underlying nature of the population's disease.
- Despite the small, heavily pretreated, molecularly unselected group of patients in this study, talazoparib demonstrated anti leukemic activity leading to stable disease and transfusion independence.
- Overall, these results suggest that further clinical trials and translational research of poly(ADP-ribose) polymerase inhibitors in hematologic malignancies are warranted to ascertain the efficacy of this class of drug in hematologic malignancies.

Supplementary data

To view the supplementary data that accompany this paper please visit the journal website at: www.futuremedicine.com/doi/suppl/10.2217/ijh-2021-0004

Author contributions

Contribution to study conception and design: GJ Mufti. Acquisition of data: RJ Mattison and M Juckett. Assisted with the data analysis: Y Chen. Data interpretation, drafting and revision of the manuscript: AK Gopal, R Popat, RJ Mattison, T Menne, A Bloor, T Gaymes, A Khwaja, M Juckett, Y Chen, M Cotter and GJ Mufti.

Financial & competing interests disclosure

This study was sponsored by Biomarin/Medivation, which was acquired by Pfizer Inc. in September 2016 (grant number not applicable). AK Gopal reports research support from Merck, Janssen, Seattle Genetics, I-Mab, IgM, Pfizer, BMS, AstraZeneca, Takeda, Spectrum, and Gilead; honoraria/consulting for Janssen, Seattle Genetics, Kite, Gilead, Amgen, Merck, Pfizer, and I-Mab. R Popat reports honoraria for Janssen, Takeda, GSK, and AbbVie; consulting for Janssen, AbbVie, and GSK; travel support from Janssen and Takeda. R Popat is supported by the National Institute for Health Research University College London Hospitals Biomedical Research Center. T Menne reports honoraria for advisory board meetings for Kite/Gilead, Amgen, Novartis, Pfizer, Celgene, Daiichi-Sankyo, and Atara; honoraria for lectures for Kite/Gilead, Takeda, Janssen, and Roche; research funding from Janssen, AstraZeneca, and Novartis. A Khwaja reports speaker fees and consulting for AbbVie, Astellas, Jazz, Novartis, and Pfizer. Y Chen is a contractor of Pfizer. MJ Cotter is an employee of Pfizer and holds Pfizer shares. GJ Mufti reports research support from BMS/Celgene and Novartis. The authors have no other relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript apart from those disclosed.

Medical writing assistance was provided by Hannah Logan, PhD, of CMC AFFINITY, McCann Health Medical Communications, and was funded by Pfizer.

Ethical conduct of research

The authors state that the study was conducted in accordance with the protocol, Good Clinical Practice standards, and the Declaration of Helsinki. The protocol was approved by the appropriate institutional review board or ethics committee at each participating institution, and all enrolled patients provided written informed consent before undergoing study-specific procedures.

Data sharing statement

Upon request, and subject to certain criteria, conditions, and exceptions (see www.pfizer.com/science/clinical-trials/trial-data-an d-results for more information), Pfizer will provide access to individual de-identified participant data from Pfizer-sponsored global interventional clinical studies conducted for medicines, vaccines, and medical devices (1) for indications that have been approved in the US and/or EU or (2) in programs that have been terminated (i.e., development for all indications has been discontinued). Pfizer will also consider requests for the protocol, data dictionary, and statistical analysis plan. Data may be requested from Pfizer trials 24 months after study completion. The de-identified participant data will be made available to researchers whose proposals meet the research criteria and other conditions, and for which an exception does not apply, via a secure portal. To gain access, data requestors must enter into a data access agreement with Pfizer.

Open access

This work is licensed under the Attribution-NonCommercial-NoDerivatives 4.0 Unported License. To view a copy of this license, visit http://creativecommons.org/licenses/by-nc-nd/4.0/

References

- 1. Rodriguez-Abreu D, Bordoni A, Zucca E. Epidemiology of hematological malignancies. Ann. Oncol. 18(Suppl. 1), i3-i8 (2007).
- Global Burden of Disease Cancer Collaboration. Global, regional, and national cancer incidence, mortality, years of life lost, years lived with disability, and disability-adjusted life-years for 29 cancer groups, 1990 to 2016: a systematic analysis for the global burden of disease study. JAMA Oncol. 4(11), 1553–1568 (2018).
- 3. Lord CJ, Ashworth A. PARP inhibitors: synthetic lethality in the clinic. Science 355(6330), 1152–1158 (2017).
- 4. Friedenson B. The BRCA1/2 pathway prevents hematologic cancers in addition to breast and ovarian cancers. *BMC Cancer* 7, 152 (2007).
- 5. Ashworth A, Lord CJ. Synthetic lethal therapies for cancer: what's next after PARP inhibitors. *Nat. Rev. Clin. Oncol.* 15(9), 564–576 (2018).
- McIornan DP, List A, Mufti GJ. Applying synthetic lethality for the selective targeting of cancer. N. Engl. J. Med. 371(18), 1725–1735 (2014).
- 7. Ribeiro HL, Soares Maia AR, Costa MB *et al.* Influence of functional polymorphisms in DNA repair genes of myelodysplastic syndrome. *Leuk. Res.* 48, 62–72 (2016).
- Janiszewska H, Bąk A, Skonieczka K *et al.* Constitutional mutations of the *CHEK2* gene are a risk factor for MDS, but not for de novo AML. *Leuk. Res.* 70, 74–78 (2018).
- 9. Ribeiro HL, Maia ARS, De Oliveira RTG *et al.* DNA repair gene expressions are related to bone marrow cellularity in myelodysplastic syndrome. *J. Clin. Pathol.* 70(11), 970–980 (2017).

- 10. Lu C, Xie M, Wendl MC *et al.* Patterns and functional implications of rare germline variants across 12 cancer types. *Nat. Commun.* 6, 10086 (2015).
- 11. Ribeiro HL, De Oliveira RT, Maia AR *et al.* Polymorphisms of DNA repair genes are related to the pathogenesis of myelodysplastic syndrome. *Hematol. Oncol.* 33(4), 220–228 (2015).
- 12. Yu G, Yin C, Wu F *et al.* Gene mutation profile and risk stratification in AML1-ETO-positive acute myeloid leukemia based on next-generation sequencing. *Oncol. Rep.* 42(6), 2333–2344 (2019).
- 13. Valka J, Vesela J, Votavova H *et al.* Genetic variant screening of DNA repair genes in myelodysplastic syndrome identifies a novel mutation in the *XRCC2* gene. *Oncol. Res. Treat.* 42(5), 263–268 (2019).
- 14. Weston VJ, Oldreive CE, Skowronska A *et al.* The PARP inhibitor olaparib induces significant killing of ATM-deficient lymphoid tumor cells in vitro and in vivo. *Blood* 116(22), 4578–4587 (2010).
- 15. Williamson CT, Muzik H, Turhan AG et al. ATM deficiency sensitizes mantle cell lymphoma cells to poly(ADP-ribose) polymerase-1 inhibitors. *Mol. Cancer Ther.* 9(2), 347–357 (2010).
- 16. Knittel G, Rehkämper T, Korovkina D *et al.* Two mouse models reveal an actionable PARP1 dependence in aggressive chronic lymphocytic leukemia. *Nat. Commun.* 8(1), 153 (2017).
- 17. Herriott A, Tudhope SJ, Junge G *et al.* PARP1 expression, activity and ex vivo sensitivity to the PARP inhibitor, talazoparib (BMN 673), in chronic lymphocytic leukemia. *Oncotarget* 6(41), 43978–43991 (2015).
- Dilley RL, Poh W, Gladstone DE et al. Poly(ADP-ribose) polymerase inhibitor CEP-8983 synergizes with bendamustine in chronic lymphocytic leukemia cells in vitro. Leuk. Res. 38(3), 411–417 (2014).
- Faraoni I, Compagnone M, Lavorgna S *et al.* BRCA1, PARP1 and γH2AX in acute myeloid leukemia: role as biomarkers of response to the PARP inhibitor olaparib. *Biochim. Biophys. Acta* 1852(3), 462–472 (2015).
- 20. Litton JK, Rugo HS, Ettl J et al. Talazoparib in patients with advanced breast cancer and a germline BRCA mutation. N. Engl. J. Med. 379(8), 753–763 (2018).
- Turner NC, Telli ML, Rugo HS et al. A Phase II study of talazoparib after platinum or cytotoxic nonplatinum regimens in patients with advanced breast cancer and germline BRCA1/2 mutations (ABRAZO). Clin. Cancer Res. 25(9), 2717–2724 (2019).
- 22. De Bono J, Ramanathan RK, Mina L *et al.* Phase I, dose-escalation, two-part trial of the PARP inhibitor talazoparib in patients with advanced germline *BRCA1/2* mutations and selected sporadic cancers. *Cancer Discov.* 7(6), 620–629 (2017).
- Murai J, Huang SY, Renaud A *et al.* Stereospecific PARP trapping by BMN 673 and comparison with olaparib and rucaparib. *Mol. Cancer Ther.* 13(2), 433–443 (2014).
- 24. Zandarashvili L, Langelier MF, Velagapudi UK *et al.* Structural basis for allosteric PARP-1 retention on DNA breaks. *Science* 368(6486), eaax6367 (2020).
- 25. European Medicines Agency. TALZENNA[®] (talazoparib) Summary of product characteristics (2019). www.ema.europa.eu/en/documents/product-information/talzenna-epar-product-information_en.pdf
- 26. Pfizer Inc. TALZENNA® (talazoparib) prescribing information (2020). http://labeling.pfizer.com/ShowLabeling.aspx?id=11046
- De Bono J, Mehra N, Higano CS *et al.* TALAPRO-1: A phase II study of talazoparib (TALA) in men with DNA damage repair mutations (DDRmut) and metastatic castration-resistant prostate cancer (mCRPC):first interim analysis (IA). *J. Clin. Oncol.* 38(Suppl. 6), 119–119 (2020).
- Agarwal N, Shore ND, Dunshee C et al. Clinical and safety outcomes of TALAPRO-2: A two-part phase III study of talazoparib (TALA) in combination with enzalutamide (ENZA) in metastatic castration-resistant prostate cancer (mCRPC). Presented at: 55th Annual Meeting of the American Society of Clinical Oncology. IL, USA (2019).
- Simon R, Rubinstein L, Arbuck SG, Christian MC, Freidlin B, Collins J. Accelerated titration designs for Phase I clinical trials in oncology. J. Natl. Cancer Inst. 89(15), 1138–1147 (1997).
- 30. Döhner H, Estey EH, Amadori S *et al.* Diagnosis and management of acute myeloid leukemia in adults: recommendations from an international expert panel, on behalf of the European leukemiaNet. *Blood* 115(3), 453–474 (2010).
- Cheson BD, Greenberg PL, Bennett JM et al. Clinical application and proposal for modification of the International Working Group (IWG) response criteria in myelodysplasia. Blood 108(2), 419–425 (2006).
- Cheson BD, Bennett JM, Kantarjian H et al. Report of an international working group to standardize response criteria for myelodysplastic syndromes. Blood 96(12), 3671–3674 (2000).
- Hallek M, Cheson BD, Catovsky D *et al.* Guidelines for the diagnosis and treatment of chronic lymphocytic leukemia: a report from the International Workshop on Chronic Lymphocytic leukemia updating the National Cancer Institute-Working Group 1996 guidelines. *Blood* 111(12), 5446–5456 (2008).
- 34. Cheson BD, Pfistner B, Juweid ME et al. Revised response criteria for malignant lymphoma. J. Clin. Oncol. 25(5), 579-586 (2007).
- 35. Yu Y, Durairaj C, Shi H, Wang DD. Population pharmacokinetics of talazoparib in patients with advanced cancer. *J. Clin. Pharmacol.* 60(2), 218–228 (2020).

- 36. Pratt G, Yap C, Oldreive C *et al.* A multi-centre phase I trial of the PARP inhibitor olaparib in patients with relapsed chronic lymphocytic leukemia, T-prolymphocytic leukemia or mantle cell lymphoma. *Br. J. Hematol.* 182(3), 429–433 (2018).
- 37. Yap TA, Plummer R, Azad NS, Helleday T. The DNA damaging revolution: PARP inhibitors and beyond. Am. Soc. Clin. Oncol. Educ. Book 39, 185–195 (2019).
- 38. Vikas P, Borcherding N, Chennamadhavuni A, Garje R. Therapeutic potential of combining PARP inhibitor and immunotherapy in solid tumors. *Front. Oncol.* 10, 570–570 (2020).
- 39. Kirtane K, Lee SJ. Racial and ethnic disparities in hematologic malignancies. Blood 130(15), 1699-1705 (2017).
- 40. Taylor J, Xiao W, Abdel-Wahab O. Diagnosis and classification of hematologic malignancies on the basis of genetics. *Blood* 130(4), 410–423 (2017).
- 41. Hanel W, Epperla N. Emerging therapies in mantle cell lymphoma. J. Hematol. Oncol. 13(1), 79 (2020).
- 42. Tseng YD, Ng AK. Hematologic malignancies. Hematol. Oncol. Clin. North Am. 34(1), 127-142 (2020).
- 43. Delia D, Mizutani S. The DNA damage response pathway in normal hematopoiesis and malignancies. *Int. J. Hematol.* 106(3), 328–334 (2017).
- 44. Beerman I, Seita J, Inlay MA, Weissman IL, Rossi DJ. Quiescent hematopoietic stem cells accumulate DNA damage during aging that is repaired upon entry into cell cycle. *Cell Stem Cell* 15(1), 37–50 (2014).
- 45. Li S, Mason CE, Melnick A. Genetic and epigenetic heterogeneity in acute myeloid leukemia. *Curr. Opin. Genet. Dev.* 36, 100–106 (2016).
- 46. Patnaik MM, Tefferi A. Cytogenetic and molecular abnormalities in chronic myelomonocytic leukemia. *Blood Cancer J.* 6(2), e393–e393 (2016).
- 47. Meggendorfer M, Jeromin S, Haferlach C, Kern W, Haferlach T. The mutational landscape of 18 investigated genes clearly separates four subtypes of myelodysplastic/myeloproliferative neoplasms. *Hematologica* 103(5), e192–e195 (2018).
- 48. Fritz C, Portwood SM, Przespolewski A, Wang ES. PARP goes the weasel! emerging role of PARP inhibitors in acute leukemias. *Blood Rev.* 45, 100696 (2021).
- 49. Robert C, Nagaria PK, Pawar N *et al.* Histone deacetylase inhibitors decrease NHEJ both by acetylation of repair factors and trapping of PARP1 at DNA double-strand breaks in chromatin. *Leuk. Res.* 45, 14–23 (2016).
- Morice PM, Leary A, Dolladille C *et al.* Myelodysplastic syndrome and acute myeloid leukemia in patients treated with PARP inhibitors: a safety meta-analysis of randomised controlled trials and a retrospective study of the WHO pharmacovigilance database. *Lancet Hematol.* 8(2), e122–e134 (2021).